Author's response to reviews

Title: Measles vaccination coverage estimates from surveys, clinic records, and immune markers in oral fluid and blood: A Population-based cross-sectional study

Authors:

  Kyla T Hayford (khayford@jhsph.edu)
  Mohammed S Shomik (mshomik@icddrb.org)
  Hassan M Al-Emran (alemranm27@hotmail.com)
  William J Moss (wmoss@jhsph.edu)
  David Bishai (dbishai@jhsph.edu)
  Orin S Levine (olevine@jhsph.edu)

Version: 6 Date: 4 December 2013

Author's response to reviews:

Dear Editorial Board of the BMC Public Health,

Please find enclosed a revised version of the manuscript: "

Based on the editor's recommendation (see below), we added two additional tables as supplementary materials and made appropriate references in the manuscript. We have made every effort to comply with the journal's style as well. Please let us know if there's anything else we can do to improve the manuscript.

Best wishes,

Kyla Hayford, on behalf of all authors.

-----

Editor's response:

"Thank you for your detailed reply. I would strongly recommend that the additional tables you provided in the response to the statistical reviewer be included as supplementary material - other readers may have the same questions. Please also provide meaningful reference to the information in the manuscript."